Author’s response to reviews

Title: The effectiveness of moxibustion for the treatment of functional constipation: a randomized, sham-controlled, patient blinded, pilot clinical trial

Authors:

Ji-Eun Park (jepark@kiom.re.kr)
Jae-Uk Sul (sjuomd@hanmail.net)
Kyungwon Kang (kwkang@kiom.re.kr)
Byung-Cheul Shin (drshinbc@gmail.com)
Kwon-Eui Hong (hkeacu@dju.ac.kr)
Sun-Mi Choi (smchoi@kiom.re.kr)

Version: 6 Date: 1 December 2011

Author’s response to reviews: see over
Answer for Editorial reviewers

Thanks for your helps and suggestions. The point to point responses to the comments of the editorial reviewer are listed below. We checked the changed sentence in red color.

Editorial reviewers' comments:

Major revisions (we require the author to make these changes)

References: please provide a functioning link to website www.cris.cdc.go.kr.
-> We added this website as reference 16. (page 17)

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
-> We removed highlighting and tracking.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements
-> I am not sure whether I understand the meaning of this sentence. If it’s possible, could you change this for me please?

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
-> We checked the figures.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
-> We checked the manuscript. Thank you.

Thank you for all valuable comments.